ADRIC: Adverse Drug Reactions In Children – a programme of research using mixed methods

Rosalind L Smyth,^{1,2*} Matthew Peak,^{3†} Mark A Turner,⁴ Anthony J Nunn,⁵ Paula R Williamson,⁶ Bridget Young,⁷ Janine Arnott,⁷ Jennifer R Bellis,³ Kim A Bird,³ Louise E Bracken,³ Elizabeth J Conroy,⁶ Lynne Cresswell,⁶ Jennifer C Duncan,³ Ruairi M Gallagher,¹ Elizabeth Gargon,⁶ Hannah Hesselgreaves,⁷ Jamie J Kirkham,⁶ Helena Mannix,³ Rebecca MD Smyth,⁸ Signe Thiesen¹ and Munir Pirmohamed⁹

¹Institute of Child Health, University of Liverpool, Liverpool, UK
²Institute of Child Health, University College London, London, UK
³Alder Hey Children's National Health Service Foundation Trust, Liverpool, UK
⁴Institute of Translational Medicine, Liverpool Women's National Health Service Foundation Trust and University of Liverpool, Liverpool, UK
⁵National Institute for Health Research Medicines for Children Research Network, University of Liverpool, Liverpool, UK
⁶Department of Biostatistics, University of Liverpool, Liverpool, UK
⁷Institute of Psychology, Health and Society, University of Liverpool, Liverpool, UK
⁸School of Nursing, Midwifery and Social Work, University of Manchester, Manchester, UK
⁹Department of Molecular and Clinical Pharmacology, University of Liverpool,

⁹Department of Molecular and Clinical Pharmacology, University of Liverpool, Liverpool, UK

*Corresponding author +Corresponding author

Declared competing interests of authors: Rosalind L Smyth has been a member of the Commission on Human Medicines and Chairperson of the Paediatric Medicines Advisory Group of the Medicines and Healthcare products Regulatory Agency (MHRA) until December 2013. Munir Pirmohamed is currently Chairperson of the Pharmacovigilance Expert Advisory Group for the MHRA and is a Commissioner on Human Medicines. Anthony J Nunn is a member of MHRA Paediatric Medicines Expert Advisory Group, a member of the European Medicines Agency Paediatric Committee (PDCO) and a temporary advisor to the World Health Organization.

Published June 2014 DOI: 10.3310/pgfar02030

Plain English summary

ADRIC: Adverse Drug Reactions In Children

Programme Grants for Applied Research 2014; Vol. 2: No. 3 DOI: 10.3310/pgfar02030

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Many common medicines have not been tested on children either properly or at all. We wanted to improve the safety of medicines used in children by investigating their side effects [adverse drug reactions (ADRs)].

We showed that three out of every 100 children admitted to hospital experienced an ADR due to a medicine taken at home. One out of five times this might have been avoidable. Nearly half of all ADRs were due to anticancer medicines.

We found that one in six children experienced an ADR while in hospital. More than half were due to medicines used in general anaesthesia and to treat pain after an operation. We used the results of these studies to develop tools to assess the likelihood of symptoms resulting from an ADR (causality) and whether an ADR was avoidable (avoidability).

We reviewed the literature on previous studies of ADRs in children and found that most studies had flaws in design and reporting, and lacked essential information, such as the name of the drugs causing ADRs and whether they were avoidable.

We interviewed children, parents and clinicians about ADRs. Many children and parents thought that communication about medicines could be improved, whereas parents of children with cancer were generally happy about the way clinicians communicated. Clinicians often made decisions about when and what to tell families but this did not always match what parents needed or expected. To address this, we developed information leaflets for parents and children to support communication between clinicians and families about ADRs.

© Queen's Printer and Controller of HMSO 2014. This work was produced by Smyth *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was set up in 2006 to produce independent research findings that will have practical application for the benefit of patients and the NHS in the relatively near future. The Programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director.

The programme is a national response mode funding scheme that aims to provide evidence to improve health outcomes in England through promotion of health, prevention of ill health, and optimal disease management (including safety and quality), with particular emphasis on conditions causing significant disease burden.

For more information about the PGfAR programme please visit the website: www.ccf.nihr.ac.uk/PGfAR

This report

The research reported in this issue of the journal was funded by PGFAR as project number RP-PG-0606-1170. The contractual start date was in August 2007. The final report began editorial review in May 2013 and was accepted for publication in January 2014. As the funder, the PGFAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGFAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2014. This work was produced by Smyth *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Programme Grants for Applied Research Editor-in-Chief

Professor Paul Little Professor of Primary Care Research, University of Southampton, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the HTA Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Professor Aileen Clarke Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Peter Davidson Director of NETSCC, HTA, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor Elaine McColl Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Faculty of Education, University of Winchester, UK

Professor Jane Norman Professor of Maternal and Fetal Health, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, University College London, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk